Author's response to reviews

Title: Androgen secreting steroid cell tumor of the ovary in a young lactating women with acute onset severe hyperandrogenism

Authors:

Altaf Gauhar Dr (altafhaji786@yahoo.com)
Shekhar Sharma Dr (dstrstrkeharsharma@gmail.com)
Vijay DK Kumar Dr (dkvijaykumar@aims.amrita.edu)
Manoj Babu Dr (manojbabur@aims.amrita.edu)
chithrathara c Dr (chithrathara@aims.amrita.edu)

Version: 6 Date: 23 October 2007

Author's response to reviews:

To

The Editor,

Subject: Article titled ¿Androgen secreting steroid cell tumor of the ovary in a young lactating women with acute onset severe hyperandrogenism¿

Sir,

Thank you for considering the said article for your journal.

In reference to the comments of the reviewers, we submit a revised version of the manuscript. A point to point reply to the suggestions is enclosed.

Referee 1:

1. Section of case report shortened and detailed description of handling of
specimen omitted.

2. Discussion revised and shortened to omit unnecessary details.

3. Figure 1 & 2 omitted.

4. Table 2 omitted.

Referee 2:

1. Grammar and syntax corrected.

2. Table 2 omitted.

Hope you find the above in order.

Regards,

Dr. Altaf Gauhar Haji, MS, DNB, Fellowship in Surgical Oncology

Fellow, Dept of Surgical Oncology,

Amrita Institute of Medical Sciences,

Cochin, India 682026